Title: Expression of a Cu,Zn superoxide dismutase typical for familial amyotrophic lateral sclerosis increases the vulnerability of neuroblastoma cells to infectious injury

Authors:

- Miriam Goos (mlotz@gwdg.de)
- Wolf-Dieter Zech (wolf.zech@web.de)
- Manoj Kumar Jaiswal (manoj.jaiswal@medizin.uni-goettingen.de)
- Saju Balakrishnan (sajbal@gmail.com)
- Sandra Ebert (sebert1@gwdg.de)
- Timothy Mitchell (T.Mitchell@bio.gla.ac.uk)
- Maria Teresa Carri (carri@uniroma2.it)
- Bernhard U. Keller (bkeller@ukps.gwdg.de)
- Roland Nau (rnau@gwdg.de)

Version: 3 Date: 7 August 2007

Author's response to reviews:

Dear Dr. Norton, dear Mrs. Brown, in July I sent you two suggestions for potential reviewers:

1. Prof. Gerald Muench, Comparative Genomics Centre, Molecular Sciences Bldg. 21, James Cook University, Townsville 4811, Australia. E-mail: Gerald.Muench@jcu.edu.au

2. Prof. Gennadij Raivich, Perinatal Brain Repair Group, Department of Obstetrics and Gynaecology, University College London, 86-96 Chenies Mews, London, UK. g.raivich@ucl.ac.uk

On your electronic submission system it says that you are still inviting reviewers. Did you get any response from these potential reviewers? Could you please send me a note how to proceed further. Sincerely yours, Roland Nau